Author's response to reviews

Title: What matters most to people with multiple myeloma? A qualitative study of views on quality of life.

Authors:

Thomas R Osborne (thomas.osborne@kcl.ac.uk)
Christina Ramsenthaler (christina.ramsenthaler@kcl.ac.uk)
Susanne de Wolf-Linder (susanne.1.de_wolf@kcl.ac.uk)
Stephen A Schey (sschey@nhs.net)
Richard J Siegert (richard.siegert@aut.ac.nz)
Polly M Edmonds (polly.edmonds@nhs.net)
Irene J Higginson (irene.higginson@kcl.ac.uk)

Version: 6 Date: 17 June 2014

Author's response to reviews:

Dear Editor,

As requested, we have confirmed in the opening sentence of the Methods that the manuscript is written in accordance with the RATS guidelines for qualitative research.

We hope this meets with your requirements, but please let us know if you have any further comments.

With warm regards,

Dr Thomas Osborne (corresponding author)
Visiting Research Fellow,
Department of Palliative Care, Policy and Rehabilitation
Cicely Saunders Institute, King’s College London
thomas.osborne@kcl.ac.uk